
Mauna Loa Solar Observatory Observer's Log

Wed Nov 11 17:06:20 GMT 1998

Year: 98 Doy: 315

Observer: koon

WEATHER COMMENT: Wed Nov 11 17:06:21 GMT 1998

Thick, dark cumulus overcast. Foggy, wet, lots of hail visible on Mauna

Kea. Wind=30 mph from the SE, temp=38 F.

MLSO PROBLEM: Wed Nov 11 17:08:06 GMT 1998

The roof of the new addition has leaked again after an apparent heavy rainstorm. Water is still dripping off the center light fixture in the computer side of the new addition, about half a gallon or so on the floor, a lot less than when Charlie finished the addition and before I tried to seal everything on the roof after the first big leak.

LOW-L COMMENT: Wed Nov 11 18:15:18 GMT 1998

L00554 popped out of drive #0, removed it and installed L00556.

PICS PROBLEM: Wed Nov 11 20:02:32 GMT 1998

I replaced the bad power supply in the camera controller. Output voltages are now good. Need good weather to test it.

WEATHER COMMENT: Wed Nov 11 20:03:18 GMT 1998

Still very cloudy, no observing.

Wed Nov 11 20:19:03 GMT 1998

MkIII